Author's response to reviews

Title: Benefits of a physician-facing tablet presentation of patient symptom data: comparing paper and electronic formats

Authors:

Daniel Glaser (daniel.s.glaser@gmail.com)
Sanjula Jain (sanjula.jain@rice.edu)
Philip Kortum (pkortum@rice.edu)

Version: 2 Date: 28 December 2012

Author's response to reviews:

December 27, 2012

To the Editor:

As requested, we have updated our paper to indicate our compliance with the Helsinki Declaration. We are pleased to resubmit this updated paper entitled, “On the benefits of a physician-facing tablet presentation of patient symptom data: comparing paper and electronic formats” to BCM Medical Informatics and Decision Making.

The following text has been added to the paper in the methods section: “The methodology employed in all of the experiments in this study were in compliance with the Helsinki Declaration and approved by Rice University’s Institutional Review Board Under FWA00003890, IRB0003061”.

Please let me know if you require any additional changes.

Sincerely,
Philip Kortum, PhD